Author's response to reviews

Title: The relationship between self-reported alcohol intake and the morbidities managed by GPs in Australia.

Authors:

Elizabeth M Proude (elizabeth.proude@usyd.edu.au)
Helena Britt (helenab@med.usyd.edu.au)
Lisa Valenti (lisav@med.usyd.edu.au)
Katherine M Conigrave (katec@med.usyd.edu.au)

Version: 5 Date: 9 February 2006

Author's response to reviews:

The title and abstract are identical in the MS and the submission system.
The authors' names are identical in the MS and the submission system, as before.
Keywords have been removed.
Box 1 has been renamed table 1; the other tables have been renumbered both in the text and in their captions. All tables are now contained in the main document, and occur after the references.
All headings are in sentence case, as before.
We have added a statement to declare that the authors have no competing interests.
Additional files have been removed.